\begin{table}%t3
\caption{Summarized stellar parameters taken from the references listed below.
The columns show the lower and upper limits of the stellar masses, radii,
ages and rotation periods. Each column is followed by a column listing the
references, which are sorted in ascending order according to the parameter
value the authors give.}\label{tab:stars2} 
%\centerline
{
\begin{tabular}{lcccccccc}
\hline\hline
      &     $M$     &     Ref.  &     $R_\star$     &     Ref.  &     Age   &     Ref.  &     $P_{\rm{rot}}$  &     Ref.  \\
      &     [$M_{\odot}$]  &           &     [$R_{\odot}$]  &           &     [Gyr] &           &     [d]   &           \\ 
      \hline
OGLE-TR-56  & 1.04  &     1  &  1.10      &     1     &           &           &           &           \\
HD 83443    & 0.79--0.98   &7, 15, 4, 2, 3      &     1.00  &     3     &     3.2--10.7    &     2     &           &           \\
HD 46375    & 0.83--1.0    &15, 5, 2, 3, 4, 6    &     1.00  &     3     &     4.5--16.5    &      2, 4  &           &           \\
HD 187123   & 0.98--1.12   &15, 8, 9, 4, 5, 2, 3   &     1.14--1.15   &      8, 3  &     4.0--10.7    &      9, 2, 5, 15   &     23--30 &      8, 9  \\
HD 179949   & 1.14--1.3    &3, 6, 5, 15   &     1.17  &     3     &     4.00  &     15    &           &           \\
HD 209458   & 1.03--1.15   &3, 15, 4, 8, 9, 2, 5      &     1.1--1.15    &      8, 3     &     2.4--5.1     &      2, 9, 4, 15   &     15.7--17     &      9, 8  \\
HD 75289    & 1.0--1.24    &3, 8, 9, 15, 2       &     1.31--1.32   &      3, 8  &     0.8--5.6     &      2, 15, 9, 2, 9, 4     &     16--17 &      9, 8  \\
BD-103166   & 0.9--1.1     &15, 10, 4    &           &           &     4.00  &     4     &         &           \\
$\tau$ Boo  & 1.3--1.42    &3, 4, 12, 8, 9       &     1.1--1.48    &      13, 3, 8, 11  &     1--2    &      4, 9, 2, 14, 9, 15    &      3.2--5.1      &       9, 8, 14, 9, 14     \\
HD 76700    & 1.00  &     3  &0.91--1.38   &      13, 3       &     4.3--7.1     &      2, 9, 4, 9, 2  &           &           \\
51 Peg      & 0.97--1.11   &15, 3, 5, 8, 9, 12, 4, 2, 11   &     0.98--1.16   &      13, 3, 8, 11  &     13.50 &     15    &     21.9--29.5   &      9, 8, 9      \\
HD 49674    & 0.95--1.0    &3, 6  &     0.95  &     3     &           &           &           &           \\
$\upsilon$ And & 1.20--1.37 & 15, 11, 12, 5, 4, 2, 8, 9, 3 & 1.2--1.69 & 13, 3, 8, 11 &
2.7--5 & 4, 9, 2, 9, 15, 14, 2 & 11.6--14 & 9, 14, 8, 9 \\ 
\hline
\end{tabular}}
\medskip
References: (1) \cite{Konacki03}, (2) \cite{Laws03}, (3) \cite{Allende99},
(4) \cite{Santos00}, (5) \cite{Santos03}, (6) Schneider, (7) \cite{Butler02},
(8) \cite{Cuntz00}, (9) \cite{Barnes01}, (10) \cite{Butler00}, (11)
\cite{Fuhrmann98}, (12)~\cite{Takeda02}, (13) \cite{CADARSJ3}, (14)
\cite{Baliunas97}, (15) \cite{Chen02}.
\end{table}